# How well is the clinical importance of study results reported? An assessment of randomized controlled trials

# Karen B.Y. Chan,<sup>\*</sup> Malcolm Man-Son-Hing,<sup>\*†</sup> Frank J. Molnar,<sup>\*†</sup> Andreas Laupacis<sup>\*‡</sup>

#### Abstract

- **Background:** The interpretation of the results of randomized controlled trials (RCTs) has traditionally emphasized statistical significance rather than clinical importance. Our aim was to assess the quality of reporting of factors related to clinical importance in a sample of published RCTs.
- **Methods:** A random sample of 27 (of a total of 266) RCTs published in 5 major medical journals over a 1-year period were reviewed by 4 independent reviewers for factors considered important in the interpretation of the clinical importance of study results: identification of a clearly defined primary outcome, reporting of the expected difference between groups used in the calculation of sample size (the delta value) and whether it was based on the minimal clinically important difference of the intervention, the statistical significance of the results, presentation of pertinent confidence intervals, and the authors' interpretation of the clinical importance of the results.
- **Results:** Twenty-two of 27 (81%) articles explicitly reported a single primary outcome. Of the 20 articles that included a sample size calculation, 18 (90%) reported a delta value. Two of the 18 (11%) articles explicitly stated that the delta value was chosen to reflect the minimal clinically important difference of the intervention. For the primary outcomes, confidence intervals surrounding the point estimates of the efficacy of the interventions were reported in 11 of 27 (41%) studies. The study results were interpreted from the perspective of clinical importance in 20 of 27 (74%) of the articles. Of these 20 reports, 5 (25%) provided justification for their clinical interpretation of the results.
- **Interpretation:** Authors of RCTs published in major general medical and internal medicine journals do not consistently provide their own interpretation of the clinical importance of their results, and they often do not provide sufficient information to allow readers to make their own interpretation.

he interpretation of the results of randomized controlled trials (RCTs) has emphasized statistical significance rather than clinical importance. For example, the revised CONSORT statement,<sup>1</sup> which is a widely adopted, recently updated series of recommendations designed to improve the quality of reporting of RCTs, failed to recommend specifically that authors discuss the clinical importance of their results. The lack of emphasis on clinical importance has led to frequent misconceptions and disagreement regarding the interpretation of the results of clinical trials and a tendency to equate statistical significance with clinical importance. In some instances, statistically significant results may not be clinically important and, conversely, statistically insignificant results do not rule out completely the possibility of clinically important effects.<sup>2,3</sup>

The minimal clinically important difference (MCID) of a therapy is defined as the smallest treatment effect that would result in a change in patient management, given its side effects, costs and inconveniences.<sup>4</sup> It is a key concept in the design of

#### Research

## Recherche

From the \*Clinical Epidemiology Unit, Ottawa Health Research Institute, Ottawa Hospital, Ottawa, Ont.; the †University of Ottawa Institute on Health of the Elderly, Sisters of Charity Health Services, Ottawa, Ont.; and the ‡Institute for Clinical Evaluative Sciences and the Department of Medicine, University of Toronto, Toronto, Ont.

This article has been peer reviewed.

CMAJ 2001;165(9):1197-202

clinical trials. Sample size calculations for prospective trials require determination of the magnitude of difference in outcomes between treatment groups that the study can reliably detect (often called the delta value). In order for clinical trials to have the best chance of detecting clinically important effect sizes, their delta values should reflect the MCIDs of the study interventions.

The MCID is also a key concept in the interpretation of clinical trial results. For example, in individuals without previous myocardial infarction (MI) or stroke, the regular use of ASA will reduce the incidence of MI by 0.2% per year (from a baseline rate of 0.7%/year to 0.5%/year, which is a relative risk reduction of about 25%), but this benefit is possibly offset by a concomitant absolute increase in the chance of stroke of 0.02% per year (from 0.30%/year to 0.32%/year, which is a relative increase of about 10%/year) and of gastrointestinal bleeding of about 1% per year (from about 1%/year to 2%/year).<sup>5</sup> After weighing the advantages and disadvantages of ASA use in this clinical situation, an influential expert panel did not recommend its use, reasoning that the efficacy of ASA in preventing MI was not sufficient to overcome the increased incidence of stroke and gastrointestinal bleeding in this low-risk group.<sup>6</sup> That is, the efficacy of ASA in this clinical situation was insufficient to meet or exceed its MCID.

The comparison of actual study results (including the point estimates and the accompanying confidence intervals) with MCID values can also give an indication of the clinical importance of the study results.<sup>7</sup> If the MCID estimate is less than the value of the lower limit of the 95% confidence interval, the study results are statistically significant and very likely to be clinically important. Alternatively, if the MCID value is greater than the upper limit of the 95% confidence interval, the results of the study are very likely to be clinically unimportant. Finally, for study results in which the MCID values are contained within the 95% confidence intervals, their clinical importance is less clear.

To allow readers to interpret the clinical importance of trial results from their own perspective, specific information must be reported, including a clearly defined primary outcome, a delta value or MCID, statistical significance and confidence intervals around the point estimates of the efficacy of the intervention. A previous review in 1987 found that such information was often underreported.<sup>8</sup> The objective of this study was to assess the quality of reporting with respect to the concept of clinical importance in a random sample of RCTs recently published in high-impact general medical and internal medicine journals.

#### Methods

For the period from Dec. 1, 1998, to Nov. 30, 1999, 5 highimpact general medical and internal medicine journals (*Annals of Internal Medicine*, *BMJ*, *JAMA*, *The New England Journal of Medicine* and the *Lancet*) were manually searched by 2 independent reviewers for RCTs. From each journal, a random sampling of 10% of the identified articles were selected for review. These articles were evaluated by 4 independent reviewers using a standardized data collection sheet developed to allow assessment of key components in the interpretation of study results from the perspective of clinical importance (Table 1). Any disagreements were resolved by collaborative review. The standardized data collection sheet evaluated the reporting of 5 factors:

- 1. Identification of a clearly defined primary outcome. The Methods section of each report was reviewed for an explicitly stated primary outcome. The primary outcome was defined as the one upon which the sample size calculation was based. If no sample size calculation was reported, the rest of the report was reviewed for an explicitly stated primary outcome. If an explicitly stated primary outcome could not be identified, the outcome that was deemed to be of greatest clinical relevance was chosen.
- 2. Reporting of the sample size delta value. Sample size calculations were evaluated for the reporting of a delta value, that is, the magnitude of difference in outcomes between treatment groups that the trial was attempting to detect. It was recorded whether the delta value was chosen to reflect the authors' perception of the MCID of the study intervention, and whether it was reported in absolute or relative terms. In order to prevent MCID values from changing with the baseline rate in the control groups (which will occur if MCID values are reported as relative risk reductions), we believe that MCID values should always be stated in absolute terms.
- 3. Documentation of the statistical significance of the results. With respect to the primary outcome, the statistical significance (p < 0.05) of the efficacy of the study intervention was recorded.
- 4. Presentation of the pertinent confidence intervals. For the primary outcome, we recorded whether the 95% confidence intervals surrounding the point estimates of the efficacy of the interventions were reported.
- 5. Authors' interpretation of the clinical importance of their results. The Discussion sections were reviewed to determine whether the clinical importance of the study results was discussed. If this was discussed, we judged whether the discussion of clinical importance was explicit or implicit. The discussion was classified as explicit if there was direct comment

#### Table 1: Methodological attributes important in the interpretation of study results from a clinical perspective

Section	Attribute	
Methods	Explicit primary outcome stated Expected magnitude of difference ("delta value") stated	
	Expected magnitude of difference identified explicitly as the MCID	
	Delta/MCID value reported as an absolute value	
Results	Statistical significance of primary outcome reported Confidence intervals for primary outcome reported	
Discussion	Clinical importance of primary outcome discussed Discussion explicit or implicit Level of justification Appropriate clinical interpretation of trial results	

Note: MCID = minimal clinically important difference.

on the clinical importance of the study results. Indirect references to clinical importance were classified as implicit. We also evaluated how well the authors justified their interpretation of the clinical importance of the study results. A rating system that graded the levels of strength of justification was developed by reviewer consensus. Levels of justification (Table 2) ranged from level 1 ("explicitly discusses the clinical importance of the primary study result in relation to previous empirical work done to determine the MCID of the therapy") to level 4 ("no accompanying justification"). Discussion of the clinical importance of study results was also reviewed for 2 other factors: justification of the magnitude of the chosen MCID value, and commentary on the relation between the MCID and the confidence intervals around the primary outcome.

## Results

From the 5 journals, 266 eligible RCTs were identified. Two reports were excluded because they presented long-term follow-up data from previously published trials. Based on a random 10% sampling of each journal's published RCTs, a total of 27 trials were selected for review.<sup>11-37</sup> The overall results of our study are shown in Table 3.

A total of 22 of 27 (81%) articles explicitly reported a single primary outcome. The remaining 5 articles measured multiple outcomes without identifying one as the primary outcome. For these articles, the primary outcome was chosen to be the one we judged most clinically relevant.

Twenty of 27 (74%) articles included a sample size calculation. Of these 20, a delta value was reported in 18 (90%). Two of the 18 (11%) articles explicitly stated that the delta value was chosen to reflect the MCID of the intervention. The delta values were reported as absolute values in 13 of 18 (72%) studies. In 2 of the 5 studies that did not report absolute delta values, the anticipated baseline event rates in the control groups were given and, therefore, absolute delta values could be calculated. Of the 27 RCTs, 17 (63%) studies had primary outcomes that were reported as being statistically significant (p < 0.05).

For the primary outcome, confidence intervals surrounding the point estimates of the efficacy of the intervention were reported in 11 of 27 (41%) of the studies.

The study results were interpreted from the perspective of clinical importance in 20 of 27 (74%) articles, with 10 of 20 (50%) articles having an explicit discussion and 10 of 20 (50%) articles having an implicit discussion.

Table 3: Reporting characteristics of published
randomized controlled trials*

Primary outcome explicitly reported	22/27 (81%)
Multiple outcomes measured without a clear primary outcome being reported	5/27 (19%)
Sample size calculation reported	20/27 (74%)
Delta value reported in sample size	
calculation	18/20 (90%)
Delta value explicitly stated as MCID	2/18 (11%)
MCID is an absolute value†	13/18 (72%)
Primary outcome statistically significant	17/27 (63%)
Confidence intervals reported for primary	
outcomes	11/27 (41%)
Clinical importance mentioned	20/27 (74%)
Explicit discussion	10/20 (50%)
Implicit discussion	10/20 (50%)
Justification	
Level 1 A–B	0/20 (0%)
Level 2	1/20 (5%)
Level 3	4/20 (20%)
Level 4	15/20 (75%)

\*Trials were published between Dec. 1, 1998, and Nov. 30, 1999, in the Annals of Internal Medicine, BMJ, JAMA, The New England Journal of Medicine and the Lancet.

†Absolute value could be calculated for 2 other studies.

Level	Justification
1A	Explicitly discusses the clinical importance of the primary study result in relation to previous empirical work done to determine the MCID of the therapy. For example, "the MCID of the CRDQ (Chronic Respiratory Disease Questionnaire) has been [empirically determined] as a change of 4 points." <sup>9</sup>
1B	Explicitly discusses the clinical importance of the primary study result in relation to an accepted, but not empirically determined, MCID for the therapy. For example, "before starting the trial, a MCID in score for each test was defined using a Delphi panel consensus." <sup>10</sup>
2	Discusses the clinical importance of the primary study result in relation to the reported sample size MCID. For example, "efficacy [did not] reach the level we initially posed as necessary to justify surgery." <sup>11</sup>
3	Discusses the clinical importance of the primary study result in relation to previously performed related studies. For example, "previous prevention interventions [have] been relatively ineffective at reducing body fatness." <sup>12</sup>
4	Mentions the clinical importance of the study result but with no accompanying justification. For example, "overall, these results provide evidence of the clinical benefits of celecoxib in the treatment of RA." <sup>13</sup>

Table 2: Levels of justification for the discussion of the clinical importance of study results

Note: RA = rheumatoid arthritis.

Most of the reports (15 of 20, 75%) provided no justification for the authors' interpretation of the clinical importance of their results (level 4). In only one article was it judged that the authors discussed the clinical importance of their study results in relation to a reported sample size MCID (level 2). No study authors justified their interpretation of clinical importance in relation to empirically determined or consensus-based MCID values (level 1). No study authors provided justification for the magnitude of their MCID value or commented on the relation of the MCID with the confidence intervals surrounding the point estimates.

#### Interpretation

The results of this study document that many reports of RCTs provide no discussion regarding the clinical importance of the trial results. These results agree with the findings of a previous review of 102 RCTs with statistically insignificant results in which only 20% (20/102) of studies made any statement about clinical importance.<sup>38</sup>

Although clinical importance was discussed in 74% (20/27) of articles in our study, we were liberal in our categorization. Several studies had only one sentence that implied clinical importance (e.g., "our findings suggest that lorazepam is safe and effective for these patients"<sup>27</sup>). For studies that did provide a discussion of clinical importance, most provided little or no justification for their clinical interpretation. The one report that ranked highest on our levels of justification (level 2) was a trial comparing adenoidectomy or tonsillectomy, or both, with medical therapy in the treatment of recurrent acute otitis media.<sup>11</sup> The sample size calculation was based on a delta value that the authors believed would represent a clinically important difference for the intervention(s). In the discussion, the authors stated, "efficacy [did not] reach the level we initially posed as necessary to justify surgery."11 This article was the only one that clearly compared the study result with an a priori MCID.

In addition to the authors' interpretation of the clinical importance of their trial results, readers should be provided with sufficient information to judge this for themselves. The study results demonstrate that many reports of RCTs did not provide the necessary information.

Eighty-one percent of the studies reviewed reported a clearly defined primary outcome. This is an improvement on a 1987 review of 45 RCTs published in 3 major medical journals in which the primary outcome measures were clearly specified in only 27% of the reports.<sup>8</sup> This trend was also apparent in the reporting of confidence intervals, with 37% of studies reviewed having reported pertinent confidence intervals compared with 13% in the previous review.<sup>8</sup>

Many studies (26%) did not report a sample size calculation or a delta value. Of those that did, many reported the delta value in relative instead of absolute terms, with an even smaller number explicitly identifying the delta value as a MCID. Furthermore, in many instances, the delta value did not appear to reflect the MCID of the intervention. For example, a study<sup>17</sup> that assessed the effect of dietary supplementation with polyunsaturated fats demonstrated an absolute decrease of 1.3% (95% confidence interval 0.1%-2.6%) in the primary outcome (combined end point of death, nonfatal MI and stroke). In the sample size calculation, the delta value was a 4% absolute difference between groups over a 3.5-year period. Thus, the efficacy of the intervention found by the study was significantly smaller than the sample size delta. If the sample size delta truly reflected the MCID of the intervention, then the efficacy of the intervention was not clinically important. However, the authors concluded that their study result provided both "a clinically important and statistically significant benefit." This issue is probably not unique to this study, because sample size calculations are often based on feasibility issues (e.g., choosing a large delta in order to reduce sample size) rather than powering the trial to have the best chance of detecting clinically important differences.<sup>39</sup> In order to change this practice with strict calculation of sample sizes based on MCIDs, feasibility issues will need to be addressed. For example, funding agencies will have to agree to provide increased support so that larger trials can be performed.

Interestingly, one trial appeared overpowered to detect the authors' perception of a clinically important benefit.<sup>15</sup> In this trial that assessed the effect of low-dose hydrocortisone on chronic fatigue syndrome, a 9-point reduction on a fatigue scale was deemed to be clinically important. However, the delta for the sample size calculation was reported as a 4-point reduction on the same scale. Thus, the authors reported that the study result, a statistically significant 4.5point reduction on the fatigue scale when comparing the intervention group with the control group, was not clinically important.<sup>15</sup>

If delta values are not representative of MCIDs and there is little discussion of the clinical importance of the study results, it is difficult to determine whether the authors perceive the results of their study to be clinically important. Previous work done by Burback and colleagues<sup>40</sup> clearly illustrates this. The clinical importance of the results of several RCTs that assessed the efficacy of tacrine in the treatment of Alzheimer's disease was unclear, because the reports provided no MCID values and the authors generally failed to comment on the clinical importance of their results. Therefore, Burback and colleagues<sup>40</sup> surveyed a group of practising physicians to determine the MCID for this intervention. The MCID value was then compared with the results of trials under review. Of the 2 of 12 studies that found a statistically significant difference in favour of tacrine therapy, neither study result could be deemed clinically important when compared with the MCID value.

One limitation of our study was that we had access only

to the final published reports of the trials we reviewed. Thus, it is possible that pertinent information regarding sample size calculations, authors' perceptions of MCID values of the interventions and interpretation of clinical importance did not survive the editorial process. In addition, our assessment of the adequacy of discussion with respect to clinical importance was subjective. Some disagreements did occur among the reviewers. However, consensus was eventually reached in all cases, with a deliberate tendency to err toward accepting ambiguous statements as indicating the presence of a discussion regarding clinical importance. Thus, our study possibly overestimated the percentage of trials that adequately addressed the issue of clinical importance. Another possible limitation is the small number of journals we reviewed.

An argument can be made that the clinical interpretation of RCTs using surrogate outcomes is problematic. However, we believe that the onus is on the authors of such trials to connect their trial results with the potential clinical implications, because not doing so leaves readers with no guidance regarding the clinical relevance of the results. With the increasing prevalence of published meta-analyses, it may be argued that the need for interpretation of the clinical importance of primary study results is superseded once a relevant meta-analysis is published. However, at the time of publication of an RCT with primary data, it is usually not known whether a meta-analysis on a similar topic will subsequently be published. Thus, the need for primary studies to publish the elements shown in Table 1 remains important. Furthermore, publication of these elements can help provide the basis with which meta-analysts (and their readers) can judge the clinical interpretation of metaanalytic results.

Our findings demonstrate that the reports of RCTs published in major general medical and internal medical journals do not provide consistently the authors' interpretation of the clinical importance of their results. Moreover, they do not contain adequate information to allow readers to assess the clinical importance of the results from their own perspective. Ideally, each published RCT should report an explicit primary outcome, a sample size delta value that reflects the authors' perception of the MCID of their intervention, the statistical significance of the primary outcome, confidence intervals surrounding the point estimates for the primary outcome and an explicit discussion of the clinical importance of the study results. Many of these criteria are already included in the revised CONSORT statement.<sup>1</sup> However, we also recommend that authors report the MCID of the intervention, justify its magnitude and explicitly discuss the clinical importance of the results in relation to the MCID.

Competing interests: None declared.

### References

- Moher D, Schulz KF, Altman DG. The CONSORT statement: revised recommendations for improving the quality of reports of parallel-group randomized trials. *Ann Intern Med* 2001;134(8):657-62.
- Sackett DL, Haynes RB, Guyatt GH, Tugwell P. Clinical epidemiology: a basic science for clinical medicine. 2nd ed. Boston: Little, Brown and Company; 1991.
- Greene WL, Concato J, Feinstein AR. Claims of equivalence in medical research: Are they supported by the evidence? *Ann Intern Med* 2000;132:715-22.
- Jaeschke R, Singer J, Guyatt GH. Measurement of health status: ascertaining the minimal clinically important difference. *Control Clin Trials* 1989;10:407-15.
- Hart RG, Halperin JL, McBride R, Benavente O, Man-Son-Hing M, Kronmal RA. Aspirin for the primary prevention of stroke and other major vascular events. *Arch Neurol* 2000;57:326-32.
- Albers GW, Easton JD, Sacco RL, Teal P. Antithrombotic and thrombolytic therapy for ischemic stroke. *Chest* 1998;114:683S-98S.
- Cook DJ, Guyatt GL, Laupacis A, Sackett DL, Goldberg RJ. Clinical recommendations using levels of evidence for antithrombotic agents. *Chest* 1995; 108:227S-30S.
- Pocock SJ, Hughes MD, Lee RJ. Statistical problems in the reporting of clinical trials: a survey of three medical journals. N Engl J Med 1987;317:426-32.
- Garrod R, Paul EA, Wedzicha JA. Supplemental oxygen during pulmonary rehabilitation in patients with COPD with exercise hypoxaemia. *Thorax* 2000;55:539-43.
- Fine MJ, Sharrock NE, Mattis S, Szatrowski TP, Charlson ME. Cognitive effects after epidural vs. general anesthesia in older adults. A randomized trial. *JAMA* 1995;274:44-50.
- Paradise JL, Bluestone CCD, Colborn DK, Bernard BS, Smith CG, Rockette HE, et al. Adenoidectomy and adenotonsillectomy for recurrent acute otitis media. Parallel randomized clinical trials in children not previously treated with tympanostomy tubes. *JAMA* 1999;282:945-53.
- Robinson TN. Reducing children's television to prevent obesity. A randomized controlled trial. *JAMA* 1999;282:1561-7.
- Simon LS, Weaver AL, Graham DY, Kivitz AJ, Lipsky PE, Hubbard RC, et al. Anti-inflammatory and upper gastrointestinal effects of celecoxib in rheumatoid arthritis: a randomized controlled trial. *JAMA* 1999;282:1921-8.
- Schofield PM, Sharples LD, Caine N, Burns S, Tait S, Wistow T, et al. Transmyocardial laser revascularization in patients with refractory angina: a randomized controlled trial. *Lancet* 1999;353:519-24.
- Cleare AJ, Heap E, Malhi GS, Wessely S, O'Keane V, Miell J. Low-dose hydrocortisone in chronic fatigue syndrome. A randomised crossover trial. *Lancet* 1999;353:455-58.
- Hansson L, Lindholm LH, Niskanen L, Lanke J, Hedner T, Niklason A, et al. Effect of angiotensin-converting enzyme inhibition compared with conventional therapy on cardiovascular morbidity and mortality in hypertension: the Captopril Prevention Project (CAPPP) randomized trial. *Lancet* 1999;353:611-6.
- GISSI-Prevenzione Investigators. Dietary supplementation with n-3 polyunsaturated fatty acids and vitamin E after myocardial infarction: results of the GISSI-Prevenzione trial. *Lancet* 1999;354(9177):447-55.
- Bo K, Talseth T, Holme I. Single blind, randomized controlled trial of pelvic floor exercises, electrical stimulation, vaginal cones, and no treatment in management of genuine stress incontinence in women. *BMJ* 1999;318:487-93.
- Bredin M, Corner J, Krishnasamy M, Plant H, Bailey C, A'Hern R. Multicentre randomized controlled trial of nursing intervention for breathlessness in patients with lung cancer. *BMJ* 1999;318(7188):901-4.
- Thomson H, Ross S, Wilson P, McConnachie A, Watson R. Randomized controlled trial of effect of Baby Check on use of health services in first 6 months of life. *BMJ* 1999;318:1740-44.
- Harland J, White M, Drinkwater C, Chinn D, Farr L, Howel D. The Newcastle exercise project: a randomized controlled trial of methods to promote physical activity in primary care. *BMJ* 1999;319:828-32.
- Hatlebakk JG, Hyggen A, Madsen PH, Walle PO, Schulz T, Mowinckel P, et al. Heartburn treatment in primary care: randomized double blind study for 8 weeks. *BMJ* 1999;319:550-3.
- Strain EC, Bigelow GE, Liebson IA, Stitzer ML. Moderate- vs high-dose methadone in the treatment of opioid dependence. A randomized trial. *JAMA* 1999;281:1000-5.
- Reynolds III CF, Frank E, Perel JM, Imber SD, Cornes C, Miller MD, et al. Nortriptyline and interpersonal psychotherapy as maintenance therapies for recurrent major depression. A randomized controlled trial in patients older than 59 years. *JAMA* 1999;281:39-45.
- Corey L, Langenberg AGM, Ashley R, Sekulovich RE, Izu AE, Douglas Jr JM. Recombinant glycoprotein vaccine for the prevention of genital HSV-2 infection. Two randomized controlled trials. *7AMA* 1999;282:331-40.
- Turner RC, Cull CA, Frighi V, Holman RR, for the UK Prospective Diabetes Study (UKPDS) Group. Glycemic control with diet, sulfonylurea, metformin or insulin in patients with Type 2 diabetes. *JAMA* 1999;281:2005-12.
   D'Onofrio G, Rathlev NK, Ulrich AS, Fish SS, Freedland ES. Lorazepam for
- D'Onofrio G, Rathlev NK, Ulrich ÁS, Fish SS, Freedland ES. Lorazepam for the prevention of recurrent seizures related to alcohol. N Engl J Med 1999; 340:915-9.

*Contributors*: All authors contributed to the study design, collected and interpreted the data, and participated in the revising of the manuscript. In addition, Drs. Man-Son-Hing and Laupacis proposed the study. Dr. Chan wrote the initial draft of the manuscript.

- Plaisance P, Lurie KG, Vicaut E, Adnet F, Petit J, Epain, et al. A comparison of standard cardiopulmonary resuscitation and active compression-decompression resuscitation for out-of-hospital cardiac arrest. N Engl J Med 1999; 341:569-75.
- Pacifico A, Hohnloser SH, Williams JH, Tao B, Saksena S, Henry PD, et al. Prevention of implantable-defibrillator shocks by treatment with sotalol. N Engl J Med 1999;340:1855-62.
- Shain RN, Piper JM, Newton ER, Perdue ST, Ramos R, Champion JD, et al. A randomized, controlled trial of a behavioral intervention to prevent sexually transmitted disease among minority women. N Engl 7 Med 1999;340:93-100.
- transmitted disease among minority women. N Engl J Med 1999;340:93-100.
  31. Cohn JN, Goldstein SO, Greenberg BH, Lorell BH, Bourge RC, Jaski BE, et al. A dose-dependent increase in mortality with vesnarinone among patients with severe heart failure. N Engl J Med 1998;339:1810-6.
- El-Sadr WM, Murphy RL, Yurik TM, Luskin-Hawk R, Cheung TW, Balfour HH, et al. Atovaquone compared with dapsone for the prevention of pneumocystis carinii pneumonia in patients with HIV infection who cannot tolerate trimethoprim, sulfonamides, or both. N Engl J Med 1998;339:1889-5.
- von Schacky C, Ångerer P, Kothny W, Theisen K, Mudra H. The effect of dietary omega-3 fatty acids on coronary atherosclerosis. A randomized, double blind, placebo-controlled trial. *Ann Intern Med* 1999;130(7):554-62.
- ICAI Study Group. Prostanoids for chronic critical leg ischemia: a randomized, controlled, open-label trial with prostaglandin E1. Ann Intern Med 1999; 130:412-21.

- Winston DJ, Pakrasi A, Busuttil RW. Prophylactic fluconazole in liver transplant recipients. A randomized, double-blind, placebo-controlled trial. Ann Intern Med 1999;131:729-37.
- Lin J, Ho H, Yu C. Chelation therapy for patients with elevated body lead burden and progressive renal insufficiency. A randomized, controlled trial. *Ann Intern Med* 999;130:7-13.
- Malmstrom K, Rodriguez-Gomez G, Guerra J, Villaran C, Pineiro A, Wei LX, et al. Oral montelukast, inhaled beclomethasone, and placebo for chronic asthma. A randomized, controlled trial. *Ann Intern Med* 1999;130:487-95.
- Moher D, Dulberg CS, Wells GA. Statistical power, sample size and their reporting in randomized controlled trials. *JAMA* 1994;272:122-4.
- 39. Feinstein AR. Clinical biostatistics. Saint Louis: CV Mosby; 1977. p. 333.
- Burback D, Molnar FJ, St John P, Man-Son-Hing M. Key methodological features of randomized controlled trials of Alzheimer's disease therapy. Minimal clinically important difference, sample size and trial duration. *Dement Geriatr Cogn Disord* 1999;10:534-40.

**Correspondence to:** Dr. M. Man-Son-Hing, Geriatric Assessment Unit, Ottawa Hospital, Civic Campus, 1053 Carling Ave., Ottawa ON K1Y 4E9; fax 613 761-5334; mhing@ottawahospital.on.ca

